# Selective Constraints in Experimentally Defined Primate Regulatory Regions

### Daniel J. Gaffney<sup>1,2</sup>\*, Ran Blekhman<sup>3</sup>, Jacek Majewski<sup>1,2</sup>

1 McGill University, Montréal, Québec, Canada, 2 Genome Québec Innovation Centre, Montréal, Québec, Canada, 3 Department of Human Genetics, University of Chicago, Chicago, Illinois, United States of America

#### Abstract

Changes in gene regulation may be important in evolution. However, the evolutionary properties of regulatory mutations are currently poorly understood. This is partly the result of an incomplete annotation of functional regulatory DNA in many species. For example, transcription factor binding sites (TFBSs), a major component of eukaryotic regulatory architecture, are typically short, degenerate, and therefore difficult to differentiate from randomly occurring, nonfunctional sequences. Furthermore, although sites such as TFBSs can be computationally predicted using evolutionary conservation as a criterion, estimates of the true level of selective constraint (defined as the fraction of strongly deleterious mutations occurring at a locus) in regulatory regions will, by definition, be upwardly biased in datasets that are a priori evolutionarily conserved. Here we investigate the fitness effects of regulatory mutations using two complementary datasets of human TFBSs that are likely to be relatively free of ascertainment bias with respect to evolutionary conservation but, importantly, are supported by experimental data. The first is a collection of almost >2,100 human TFBSs drawn from the literature in the TRANSFAC database, and the second is derived from several recent high-throughput chromatin immunoprecipitation coupled with genomic microarray (ChIP-chip) analyses. We also define a set of putative cis-regulatory modules (pCRMs) by spatially clustering multiple TFBSs that regulate the same gene. We find that a relatively high proportion ( $\sim$ 37%) of mutations at TFBSs are strongly deleterious, similar to that at a 2-fold degenerate protein-coding site. However, constraint is significantly reduced in human and chimpanzee pCRMS and ChIP-chip sequences, relative to macaques. We estimate that the fraction of regulatory mutations that have been driven to fixation by positive selection in humans is not significantly different from zero. We also find that the level of selective constraint in our TFBSs, pCRMs, and ChIP-chip sequences is negatively correlated with the expression breadth of the regulated gene, whereas the opposite relationship holds at that gene's nonsynonymous and synonymous sites. Finally, we find that the rate of protein evolution in a transcription factor appears to be positively correlated with the breadth of expression of the gene it regulates. Our study suggests that strongly deleterious regulatory mutations are considerably more likely (1.6-fold) to occur in tissue-specific than in housekeeping genes, implying that there is a fitness cost to increasing "complexity" of gene expression.

Citation: Gaffney DJ, Blekhman R, Majewski J (2008) Selective Constraints in Experimentally Defined Primate Regulatory Regions. PLoS Genet 4(8): e1000157. doi:10.1371/journal.pgen.1000157

Editor: Emmanouil T. Dermitzakis, Wellcome Trust Sanger Institute, United Kingdom

Received December 19, 2007; Accepted July 9, 2008; Published August 15, 2008

**Copyright:** © 2008 Gaffney et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: DJG was funded by FRSQ postdoctoral fellowship 13389 and by CIHR grant 209549 to JM. RB is supported by NIH grant GM077959 to Yoav Gilad.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: daniel.gaffney@mcgill.ca

#### Introduction

Changes in gene regulation are likely to play an important role in evolution [1,2]. However, compared to protein-coding sequences, the fitness effects of regulatory mutations remain poorly understood. Furthermore, the relationship between changes in gene regulatory regions and the expression phenotype of the regulated gene are unclear. Both of these issues are partly a result of poor annotation of the sites that control gene regulation, the vast majority of which are likely to be noncoding. For example, transcription factor binding sites (TFBSs), a major component of regulatory architecture, are small (6–15 bp), laborious to identify experimentally and potentially degenerate. Furthermore, due to their small size, genuine TFBS are difficult to differentiate from similar, randomly-occurring sequences that are present in large numbers in mammalian genomes.

In an attempt to address the problem of annotation, evolutionary conservation has become popular as a metric for identifying putative regulatory regions [3]. However estimates of the true level of selective constraint (defined as the proportion of mutations which are strongly deleterious) in regulatory DNA will, by definition, be biased upwards in datasets predicted using evolutionary conservation as a criterion.

One way to address this problem is to focus solely on regulatory regions which have been defined primarily by experimental rather than evolutionary criteria. In this study, we estimated levels of selective constraint in mammalian regulatory noncoding DNA using two complementary datasets, both of which draw upon experimental data. The first was derived from the literature collected in the TRANSFAC database [4], such that every TFBS is supported by at least a single refereed publication. The advantages of this dataset are twofold. First, our dataset consists of individual TFBSs for which experimental support exists and which, according to an analysis by publication date (see Discussion), appear to be subject to relatively little ascertainment bias with respect to evolutionary conservation. Second, the literature in TRANSFAC also provides substantial information on the gene regulated and the transcription factor bound for each

#### Author Summary

Changes in gene expression have been suggested to play a major role in mammalian evolution. In eukaryotes, gene expression is primarily controlled by sites, such as transcription factor binding sites (TFBSs), located in the noncoding region of the genome. The majority of these TFBSs remain unannotated, however, because they are typically short, degenerate, and laborious to identify experimentally. As a result, the effects of mutations in TFBSs on organism fitness remain poorly understood. We collected a dataset of TFBSs derived from the experimental biology literature and recent high-throughput studies to estimate the proportions of new mutations in TFBSs that have strongly deleterious and strongly beneficial effects upon organism fitness. We find that a relatively high proportion of new mutations in TFBSs are strongly deleterious, although it appears that relatively few are adaptive. We also demonstrate that the fraction of strongly deleterious regulatory mutations is correlated with the breadth of expression of the regulated gene. Thus, ubiquitously expressed genes are likely to experience fewer deleterious regulatory mutations than those expressed in a small number of tissues.

TFBS. Thus, the TFBSs in our dataset can be assigned to a specific gene reliably, and we can also determine at least some of the transcription factors (TFs) which regulate a specific gene's expression. Our second dataset comprises sequences which have been identified as potentially transcription-factor-binding using chromatin immunoprecipitation combined with genomic micro-array (ChIP-chip) analyses. Specifically, we combine the sequences annotated in refs 5–11. While the resolution at which regulatory sites are identified is undoubtedly lower in the ChIP-chip dataset than in our TFBS dataset, our ChIP-chip sequences will still be highly enriched for functional regulatory DNA.

Using these two datasets, we addressed the following questions: (i) what fraction of regulatory mutations in primates are strongly deleterious, (ii) does the fraction of strongly deleterious mutations at TFBSs vary between primates, (iii) what fraction of substitutions in human regulatory regions have been driven to fixation by positive selection, (iv) how does the selective constraint of human regulatory noncoding regions relate to the expression profile of the gene they regulate and (v) does the rate of protein evolution of a TFs also relate to the expression profile of the regulated gene?

#### Results

Of the 2494 human TFBS accessions in TRANSFAC 10.2 we were able to successfully locate a total of 2097 TFBSs on human genome assembly 18. Many of these regions overlapped and so we were able to define a total of 1508 unique TFBSs, corresponding to over 18 kb of sequence in the human lineage. The TFBSs in our study were a highly heterogeneous mix of noncoding binding regions which included TATA boxes, CCAAT/enhancer binding protein sites (C/EBP sites), cAMP response elements (CREs), interferon-stimulated response elements (ISREs) as well as binding sites for a variety of common TFs such as cMyc, Sp1, NF-κB, CREB and p53. Those 1355 TFBSs for which we were able to extract orthologous sequence from both the chimpanzee and macaque genomes were grouped according to location into a total of 468 case regions, as described in the Methods.

Of the 12364 ChIP-chip sequences annotated in refs 5–11 we defined 10104 unique, nonoverlapping regions corresponding to

5.3Mb of human sequence which were divided into 6712 "case" regions. The genomic locations of the TFBSs and ChIP-chip sequences are shown in Figure 1.

Using parsimony, we estimated rates of nucleotide substitution at non-CpG-prone sites in the TFBSs, their flanking sequences, the ChIP-chip sequences and our neutral control regions. Both TFBSs and ChIP-chip sequences are evolving significantly slower than their respective controls (Figure S1). This suggests that our controls contain significantly more neutrally evolving sites than both our regions of interest and likely provide a reasonable estimate of the local mutation rate, a major assumption of our method.

Individual TFBSs are highly non-randomly spatially distributed in metazoans, and are typically located within clusters of other binding sites termed cis-regulatory modules (CRMs). These clusters serve to bind multiple TFs whose combined action controls the level and location of gene expression. We investigated whether this modular organisation of regulatory DNA was reflected by the level of selective constraint in the noncoding DNA surrounding known TFBSs. In order to distinguish between regions within CRMs versus those at the outermost edge we divided our dataset of flanking sequences into two groups: (i) those flanking regions that lie between two annotated TRANSFAC binding sites which are less than 1.5 kb apart and (ii) those flanking regions that are greater than 1.5 kb from another annotated binding site, coding sequence or TSS. These criteria will enrich group 1 in sequence lying within a CRM, while those in group 2 will be enriched in sequence lying at the edge of a CRM. It is clear from Figures 2 and 3 that sequence adjacent to our TFBSs is selectively constrained for a considerable distance upstream and downstream of the binding site itself, suggesting that our annotated TFBSs are likely surrounded by other regulatory regions. Furthermore many of the sites which, according to their TRANSFAC annotation, co-regulate the same gene, are located in close proximity indicating that they function co-cooperatively as a CRM. We therefore defined a set of 696 putative CRMs (pCRMs) in our dataset as all TFBSs which regulated the same gene and were <350 bp from one another, their intervening sequence and up to 350 bp of flanking sequence. The mean length of a pCRM in our data was 415 bp.

## Constraint in Primate TFBSs, pCRMs, and ChIP-chip Sequences

We next estimated the level of selective constraint at TFBSs, pCRMs and in ChIP-chip sequences (Figure 4). TFBSs appear to be reasonably highly constrained, approximately equivalent to a 2-fold degenerate synonymous site. This result is in good agreement with previous studies which have suggested that a reasonable proportion of TRANSFAC binding sites are conserved between human and a variety of mammalian species [12–16]. Estimates of constraint in our putative *cis*-regulatory modules and sequences annotated by ChIP-chip experiments are somewhat similar (0.14 and 0.11, respectively) suggesting that ChIP-chip studies can serve as reliable guides to functional regulatory regions in humans when compared with more traditional methods of identification.

#### Variation in Selective Constraint between Primates

It has been suggested that regulatory DNA in primates is under relaxed selective constraint relative to rodents [17]. This has been attributed to the reduction of effective population size in primates facilitating the fixation of slightly deleterious mutations in gene control regions. Effective population size is likely to vary between humans, chimpanzees and macaques and we therefore investigated whether any significant difference in constraint of regulatory noncoding regions existed between humans and their close



Figure 1. Proportion of the total number of sites contributed by TFBSs (A) and ChIP-chip sequences (C) in different genomic regions, and frequency distribution of the distance of TFBS (B) or ChIP-chip sequence (D) from the transcription start site (TSS) of the regulated gene. doi:10.1371/journal.pgen.1000157.g001

relatives. It is clear from Figure 5 that selective constraints in regulatory noncoding DNA vary significantly between primate species (1-way ANOVA,  $P < 10^{-16}$ ), and that this is primarily a result of a reduction in constraint in hominins (*post-hoc* Tukey test,  $P < 10^{-5}$  human vs rhesus, chimp vs rhesus) when compared with rhesus macaques. Summing over both coding and noncoding sites, we also find that mean selective constraint is also reduced somewhat in humans, compared with chimpanzees (0.254 versus 0.279), although this difference is only marginally significant (Bootstrap t-test, P < 0.07). Two possible explanations for the reduction in constraint are that humans and chimpanzees have accumulated substantially greater numbers of deleterious mutations or are experiencing higher rates of adaptive evolution in their regulatory regions.

#### Adaptive Evolution in Primate Regulatory DNA

In order to investigate whether the reduced constraints we observed in human regulatory noncoding DNA were the result of adaptive evolution we estimated the proportion of substitutions ( $\alpha$ ) which were driven to fixation by positive selection in our pCRMs and ChIP-chip sequences using the McDonald-Kreitman framework [18,19]. We were able to map 232 of our pCRMs and ChIP-chip sequences onto regions sequenced by the NIEHS Environmental Genome Project (EGP; http://egp.gs.washington.edu).

Polymorphism data was taken from the EGP as this dataset is free of ascertainment bias, relative to other large polymorphism datasets, such as HapMap [20]. McDonald-Kreitman analyses assumes that all mutations can be divided into strongly selected (positively or negatively) or strictly neutral classes. One problem with this is that a non-negligible fraction of new mutations in species with small effective population sizes, such as primates, may be weakly negatively selected. To account for this possibility we estimated  $\alpha$  using both all segregating sites and excluding those sites where the minor allele frequency MAF ranged from 0.01 to 0.30, many of which are likely to be slightly deleterious [21]. We find no evidence of adaptive evolution in human regulatory regions and our estimate of  $\alpha$  is not significantly different from zero across the entire range of excluded, low frequency polymorphisms (Figure 6).

## Selective Constraint in Regulatory Noncoding DNA and Gene Expression Profile

We next investigated whether constraint in our regulatory sites covaried with the expression breadth of the gene regulated. Expression profile of the genes inferred to be regulated by our pCRMs and ChIP-chip sequences was estimated from the human microarray data of Su *et al* [22]. A gene was defined as expressed in a specific tissue based on the Affymetrix MAS5 presence/absence



**Figure 2. Selective constraint of flanking sequence located between two annotated TFBSs that are** <1.5 kb apart. Dotted lines show 95% confidence intervals estimated by bootstrapping the data by case-control region, 1000 times. doi:10.1371/journal.pgen.1000157.g002



Figure 3. Selective constraint of flanking sequences for which the annotated TFBS was >1.5 kb from another annotated TFBS, coding sequence or TSS. Dotted lines show 95% confidence intervals estimated by bootstrapping the data by case-control region, 1000 times.

doi:10.1371/journal.pgen.1000157.g003

calls. Our results were qualitatively unchanged when gene expression was designated using a cutoff probe intensity value (data not shown). For comparison, we also estimated constraint at the nonsynonymous (0-fold degenerate) and synonymous sites of the genes adjacent to our regulatory regions. The results of this analysis are presented in Figure 7. There is a clear relationship between selective constraint in regulatory regions and breadth of expression of the regulated gene. pCRM and ChIP-chip selective constraint is significantly negatively correlated with the number of tissues in which a gene is expressed (P<0.005, P< $5.07 \times 10^{-7}$ ,



Figure 4. Estimates of selective constraint at TFBSs, pCRMs and ChIP-chip sequences averaged across all three primates and 4fold, 2-fold and 0-fold degenerate sites. Genes used were those regulated by the TFBSs and inferred to be regulated by the ChIP-chip sequences.

doi:10.1371/journal.pgen.1000157.g004



Figure 5. Selective constraint of regulatory noncoding (TFBSs, pCRMs and ChIP-chip sequences) and coding (nonsynonymous sites) DNA in humans, chimpanzees and macaques. doi:10.1371/journal.pgen.1000157.g005

respectively). This is not a function of the number of annotated TFBSs in our pCRMs, which is uncorrelated (Pearson r = 0.015; P<0.738) with pCRM constraint. A similar relationship appears to exist between TFBS selective constraint and expression breadth. In particular, the TFBSs of tissue-specific genes are more highly constrained than those of intermediate and broadly expressed genes (2-sided t-test; P<0.012). However, the equivalent regression is not significant at least in part due to the high error involved in estimating selective constraint from a small number of sites between closely related species. The relationship between constraint and expression profile is reversed in protein-coding sequence where constraint increases with increasing expression breadth (P<1.11×10<sup>-15</sup> and P<1.70×10<sup>-6</sup>, nonsynonymous and



Figure 6. Fraction of adaptive substitutions ( $\alpha$ ) in primate pCRMs and ChIP-chip sequences versus the threshold minor allele frequency (MAF) that was excluded from the analysis prior to the estimation of  $\alpha$  (see text). Confidence intervals are shown as dashed lines and were estimated by bootstrapping the data by case-control region, 10000 times. doi:10.1371/journal.pgen.1000157.g006



2-fold degenerate, respectively), a result supported by previous work [23,24,25]. Interestingly, constraint at 4-fold degenerate synonymous sites is also positively correlated with expression breadth ( $P < 2.60 \times 10^{-6}$ ), suggesting that constraints on mRNA stability and/or splicing efficiency reflect those on protein structure, with respect to expression breadth. These results are not a product of different rates of nucleotide substitution in the intronic control regions of genes with differing expression breadth; we find that divergence in all controls used in our study was uncorrelated with breadth of expression of the gene in which they reside (Pearson r = -0.004 P > 0.83; Figure S2).

It has previously been shown that mammalian promoters can be divided into two classes, CpG-rich and CpG-poor, based on the distribution of %CpG in human promoter regions [26] and these two classes of promoter region are associated with expression breadth. Following ref 26, we divided our pCRM and ChIP-chip sequences into CpG-rich and CpG-poor classes, to investigate whether this could explain the relationship we find between expression breadth and conservation. The majority (95%) of our pCRMs and ChIP-chip sequences are CpG-rich by the definition in ref 26 i.e. they have a normalized CpG content of >0.35. Within this CpG-rich class, constraint of regulatory regions is still significantly negatively correlated with expression breadth (Pearson r = -0.103,  $P < 3.68 \times 10^{-8}$ ). We also tested the influence of CpG content by regressing pCRM and ChIP-chip constraint on their %CpG. The slope of this regression is negative and significantly different from zero (simple linear regression b = -0.058, P<0.028). However, the residuals of this regression

Figure 7. Constraint in regulatory noncoding DNA (TFBSs, pCRMs, ChIP-chip sequences) and coding regions (nonsynonymous, 2-fold and 4-fold synonymous sites) versus gene expression breadth. Narrow, intermediate and broad expression breadth were defined using lower (2 tissues) and upper quartiles (>30 tissues) of the distribution of number of tissues expressed per gene. Constraint at pCRMs, ChIP-chip sequences, nonsynonymous, 2-fold and 4-fold degenerate sites was significantly correlated with number of tissues in which a gene was expressed (Pearson r = -0.144, P<0.005; r = -0.092, <5.07×10<sup>-7</sup>; r = 0.176, P<1.22×10<sup>-12</sup>; r = 0.099, P<6.63×10<sup>-5</sup>; r = 0.110, P<9.36×10<sup>-6</sup>, respectively). doi:10.1371/journal.pgen.1000157.g007

are still negatively correlated with expression breadth (Pearson r = -0.086,  $P < 5.99 \times 10^{-7}$ ). These results suggest that, while CpG content is indeed correlated with constraint of regulatory DNA this does not explain the majority of the relationship we see between regulatory constraint and expression profile.

#### Transcription Factor Dn/Ds and Gene Expression Profile

One advantage of our TFBS dataset is that we can identify which TF(s) control the expression of a specific gene, and that this relationship is also supported by experimental evidence. We therefore investigated whether the rate of protein evolution (estimated as Dn/Ds, the ratio of nonsynonymous to synonymous substitution) in a TF bore any relationship to the expression breadth of the regulated gene. Dn/Ds was estimated summing over all sites of all TFs which were known to regulate a specific gene. We obtained Dn/Ds estimates for 185 TFs which regulate 349 genes. The results of this analysis are presented in Figure 8. Interestingly we find that TF Dn/Ds ratio is significantly positively correlated with gene expression breadth (Pearson r = 0.15; P<0.005). We tested whether this result was an artifact of summing across multiple TFs by restricting our analysis to the 99 genes which were regulated by a single TF. Despite this reduced dataset TF Dn/Ds is still marginally significantly correlated with gene expression breadth (P<0.076).

One major factor which influences the rate of protein evolution is their structure. We therefore tested whether the relationship between transcription factor Dn/Ds and gene expression profile was influenced by the transcription factor structural class. We divided the regulating TFs into four protein "superclasses" based on the transcription factor protein classification tree in TRANS-FAC. Of our 185 TFs we were able to assign 141 to either leucine zipper factors (LZ; 27 TFs), zinc-coordinating DNA-binding domains (ZC; 50 TFs), helix-turn-helix proteins (HTH; 42 TFs) and  $\beta$ -scaffold factors with minor groove contacts (BSF; 22 TFs). As expected, we find clear differences in the Dn/Ds ratio of each the four classes (1-way ANOVA P $\ll$ 1.69 $\times$ 10<sup>-5</sup>; Figure S3). However, we find no relationship between protein structural class and expression breadth of the regulated gene (1-way ANOVA P<0.464; Figure S4). This suggests that the relationship we



Figure 8. Dn/Ds of the transcription factors (TFs) regulating gene expression versus gene expression breadth. Dn/Ds was estimated from human-macaque alignments, treating all TFs known to regulate each gene as a single sequence. doi:10.1371/journal.pgen.1000157.g008

observe between the rate of protein evolution in a TF and the expression breadth profile of the regulated gene are independent of the protein structure of the TF.

#### Discussion

We have presented a study of the fitness effects of mutations in primate regulatory noncoding DNA. The regulatory regions included in our study are supported by a variety of experimental sources, both based on the extensive experimental biology literature, and inferred from more recent, high-throughput studies. Our study confirms that experimentally validated regulatory noncoding regions are selectively constrained, a result supported by other previous studies of datasets of TRANSFAC TFBSs in mammals [12-16]. Our estimates imply that  $\sim 37\%$  of new spontaneous mutations in primate TFBSs have a strongly deleterious effect and are removed by purifying selection. We find that the proportion of strongly deleterious noncoding regulatory mutations varies significantly even between closely-related primate species, reflecting a similar trend in coding DNA. We find no evidence for adaptive evolution in human regulatory regions, suggesting that these differences in selective constraint between primate taxa are likely to primarily reflect variations in effective population size. Our study also clearly shows that the level of selective constraint in primate regulatory DNA depends upon the expression profile of the gene regulated. Intriguingly, we also find higher constraint in the regulatory regions of tissue-specific genes is reflected in the rate of protein evolution of the TFs that interact with them.

Our study suggests that at least some fraction of human regulatory DNA is accumulating slightly deleterious mutations at an accelerated rate relative to other, closely-related primate species. We find no evidence of adaptive evolution in our regulatory regions. Nonetheless, a number of recent reports have suggested accelerated evolution in human noncoding DNA [27,28,29]. There may be a number of reasons that we do not observe such an effect. Firstly, we restrict our analysis to experimentally-supported regulatory noncoding DNA and exclude CpG prone sites entirely from our analysis and may therefore lack sufficient power to detect all but very strong selection. Secondly our analysis is based upon the McDonald-Kreitman test which assumes that all adaptive mutations are strongly selected. However, recent work has suggested that at least some fraction of adaptive mutations may be weakly selected [30]. Although our degree of confidence in our estimates of  $\alpha$  is small, the increasing numbers of high quality ChIP-chip datasets combined with larger resequencing studies will improve the accuracy of estimates of this important parameter.

The results we have presented also shed light on the relationship between gene expression and selective constraint of both the TF and TFBSs which ultimately control this expression. A straightforward interpretation of our results is that selective constraint of regulatory DNA parallels the "complexity" of expression of the gene it regulates i.e. genes that are required to be "switched on" ubiquitously have a simpler, more degenerate regulatory architecture than those genes which require delicate control of the location and timing of expression. This interpretation is supported by a recent study of human-mouse promoter regions [31]. Furthermore, this hypothesis is intuitively appealing when we consider that tissue-specific genes may require regulatory sites both to up-regulate expression in the correct tissue, but also to suppress expression in an inappropriate tissue, a function that is presumably absent from the regulatory region of a broadly-expressed gene. Taken together with estimates of constraint in protein-coding sequence our study suggests the following: broadly expressed genes produce a protein whose structure is tightly maintained by

purifying selection but whose regulatory architecture is degenerate. Tissue-specific genes on the other hand require a more elaborate and specific regulatory apparatus, but the protein produced by such genes is less rigorously maintained by selection. It has been suggested that mutations affecting the regulation of tissue-specific genes are less likely to be strongly deleterious than those in broadly-expressed genes, given that they are expressed in a subset of tissues [32]. However, our results support the opposite interpretation.

Although the correlations we observe between regulatory constraint and expression breadth are weak, we note that the experimental methods of annotation of regulatory sites are imperfect, and the numbers of sites which we have used in this study are relatively small, by genomic standards. In addition, estimates of selective constraint are essentially a ratio of ratios, making them inherently noisy. In the light of this, the strength of our correlations is perhaps less surprising. It is also likely that our results to a certain extent reflect the variation in constraint of the noncoding DNA surrounding different functional "classes" of genes, as demonstrated previously (e.g. ref 33). We note, however, that the relationship between gene expression profile and gene functional class as assigned by ontological classification is uncertain. In addition, without a complete annotation of functional noncoding sites, we cannot distinguish whether between-gene variation in constraint of surrounding noncoding regions reflects variation in the number of constrained sites or in the *intensity of purifying selection* at these sites. One advantage of the approach we have employed here is that we can at least partially disentangle these two factors; our results suggest that the intensity of purifying selection at primate TFBSs is indeed greater in tissue-specific genes (Figure 7).

We note that our estimates of constraint may also be biased upwards for two reasons. The UCSC whole genome alignments are assembled with reference to the human genome and it is therefore possible that their use in our study could exclude weakly conserved unalignable TFBSs. This could potentially lead to an overestimate of the true level of constraint. However, we suggest that the impact of this is likely to be small given that such a bias will affect our control regions also, and thus will cancel in the estimation of constraint. In addition, although ascertainment bias in the TRANSFAC annotations is reduced, compared to some computationally-predicted regions, it is unlikely to be zero, as phylogenetic footprinting has become more frequently used over time as a means of selecting candidate regulatory regions for experimental testing. Unfortunately it is difficult to quantify this bias. However, if phylogenetic footprinting has had a significant effect upon our TFBS dataset we might predict that, on average, those TFBSs that were annotated relatively recently would be less diverged than those annotated in the more distant past, given the dramatic increase in the use of comparative genomics in recent years. We find that divergence is not significantly correlated with year of appearence of the supporting publication (Figure S5). We do find that TFBSs published before 1996 (the median age of publication of human TRANSFAC TFBSs) are marginally (~8%) more diverged than those published during or after 1996, although this difference is not significant (Bootstrap t-test, P < 0.19). Thus, although our estimates of TFBS constraint may be upwardly biased, this bias is likely to be small.

One straightforward implication of our results is that deleterious regulatory mutations are more likely to disrupt genes with tissue-specific expression, as a result of higher levels of constraint in both their regulatory regions and the protein-coding sequence of the TFs that bind to these regions. We estimate that deleterious mutations will occur on average 1.6-fold more often in regulatory regions of tissue-specific ( $\leq 3$  tissues) than housekeeping genes

(>35 tissues). This conclusion has interesting implications when we consider recent evidence suggesting that there are substantially more tissue-specific genes in primates compared with rodents [34]. Our data imply that the penalty for an increase in expression "complexity" is a concurrent increase in the genomic deleterious mutation rate. This penalty may, however, be offset by a corresponding decrease in the proportion of deleterious proteincoding mutations.

#### **Materials and Methods**

#### Data Collection

The data used in this study were collected from two sources. We first used the literature in TRANSFAC release 10.2 [4] to compile a dataset of known, experimentally-supported TFBSs. For those TFBSs which were linked to a specific EMBL accession, we BLASTed the binding site and up to 400 bp of flanking sequence against the human genome (assembly 18). Query sequences which matched a single, unique region in the human genome with a BLAST *e*-value of  $<10^{-5}$  were accepted. Those regions which matched more than a single location were resolved manually by comparison with any existing annotation in TRANSFAC, or excluded. For those TFBSs which were not linked to an existing EMBL record, we BLASTed the binding site sequence against the transcript of the RefSeq gene regulated, as recorded in TRANSFAC, with 20 kb flanking sequence. We accepted any binding site which matched a single unique location in this sequence, with <99% identity, for the full length of the binding site. We hereafter refer to these data as "TFBS" sequences. All binding sites were checked to be in the appropriate chromosomal location with respect to the gene they regulate. Our second dataset was derived from DNA sequences bound by a variety of TFs in 7 recent chromatin immunoprecipitation-coupled DNA microarray (ChIP-on-chip) analyses [5-11]. The locations of these sequences were extracted from the "fragment" table of TRANSFAC 10.2 and updated to the latest assembly of the human genome. We hereafter refer to these data as "ChIP-chip" sequences.

To estimate the level of selective constraint, we needed to compare substitution rates in our TFBS and ChIP-chip datasets with those in an appropriate neutrally-evolving control region, which has a mutation rate equal to that of the region of interest. Previous analyses [33,35] have suggested that, in mammals, intronic regions outside the first intron and the splice sites are the fastest evolving in the genome and among the best candidates for neutrally-evolving sequence. Because sites in both datasets were highly nonrandomly distributed across the genome, we sought to define a single control region for a "case" region of binding sites, rather than for each individual annotated sequence. A "case" was defined as a group of TFBSs or ChIP-chip sequences in which the maximum distance between each cluster member and its nearest neighbour was 100 kb. A control region for each "case" was defined as the window which extended up to 250 kb either side of the midpoint of cluster. All non-first intronic sequence, excluding the first and last 100 bp, within this 500 kb window were denoted as control regions for the "case" region. Given that mutation rates in mammals appear to vary across megabase scales [36,37] it is likely that the mutation rate in our control sites will not differ significantly from that in our "case" sites. In a minority of cases (<5% of TFBSs and ChIP-chip sequences), suitable intronic controls were unavailable. In this case, we used nearby intergenic sequence which was greater than 1 kb from an annotated coding sequence. All exon locations were taken from RefSeq annotations. The selection of an arbitrary between-site distance of 100 kb allowed us to define 473 unique, nonoverlapping binding site

"cases", each with a unique set of intronic controls. Likewise, we defined 6712 ChIP-chip "cases" from our 10104 unique ChIP-chip regions.

#### Analysis

For all TFBSs, ChIP-chip sequences and their corresponding control regions, aligned sequence data from the human, chimpanzee (assembly 2) and macaque (assembly 2) genomes was extracted from the 28-way vertebrate alignments available in the UCSC genome browser database [38]. In order to minimize the effects of poor sequence quality in the chimp and macaque genomes we masked all sites which were assigned a base quality of less than 20 in either species. Lineage-specific substitution rates were estimated using parsimony. Estimates of substitution rates were not corrected for multiple hits, given that this will make little difference between closely related species.

In all cases, selective constraint, C, was estimated as:

$$C = 1 - O/E$$

where O is the number of substitutions observed in the TFBS or other region of interest and E is the number of substitutions expected under neutral evolution:

$$E = n \times K$$

where n is the length of the TFBS or other region of interest and K is the substitution rate estimated from the control region. Unless stated otherwise, all confidence intervals were estimated by bootstrapping the data by binding site "case", 1000 times.

The method of estimation of selective constraint employed here explicitly accounts for local mutational variation. Previous studies of experimentally validated mammalian regulatory DNA (e.g. refs 14–16), have not accounted for such variation. This is particularly important in our study for two reasons. Firstly, substantial withingenome mutational variation is known to occur in mammals [37,39] meaning that the expectation of conservation under neutrality will vary from one genomic region to the next. This can substantially impact estimates of conservation between very closely related species, such as humans and chimpanzees. Secondly, regulatory regions frequently reside in CpG islands, where the level of CpG hypermutability is known to differ from other, more heavily methylated regions of the genome. Given that CpG mutations make up a disproportionately large number of all mutations in mammals, it is important to correct for variations in the level of CpG hypermutability to avoid overestimating constraint in regions of lowered CpG hypermutability such as CpG islands. Here, we account for variation in the frequency and mutability of CpG dinucleotides by excluding non CpG-prone sites (not preceded by 'C' or followed by 'G').

#### References

- King MC, Wilson A (1975) Evolution at two levels in humans and chimpanzees. Science 188: 107–116.
- Carroll SB (2005) Evolution at two levels: On genes and form. PLoS Biology 3: 1159–1166.
- Blanchette M, Bataille AR, Chen XY, Poitras C, Laganiere J, et al. (2006) Genome-wide computational prediction of transcriptional regulatory modules reveals new insights into human gene expression. Genome Res 16: 656–668.
- Wingender E, Chen X, Hehl R, Karas H, Liebich I, et al. (2000) TRANSFAC: an integrated system for gene expression regulation. Nucleic Acids Research 28: 316–319.
- Kapranov P, Cawley SE, Drenkow J, Bekiranov S, Strausberg RL, et al. (2002) Large-scale transcriptional activity in chromosomes 21 and 22. Science 296: 916–919.

TF Dn/Ds ratios were estimated from human-macaque alignments in the Cornell orthologues dataset using PAML [40]. In the case where multiple factors were known to regulate a gene *x*, Dn/Ds ( $\omega_x$ ) was estimated summing over all TFs, T<sub>x</sub> =  $t_1,...,t_n$  as:

$$\omega_x = \frac{\sum_{T_x} K_A(t_i)}{\sum_{T_x} N_A(t_i)} / \frac{\sum_{T_x} K_S(t_i)}{\sum_{T_x} N_S(t_i)}$$

where  $K_A(t_i)$  and  $K_s(t_i)$  are the number of pairwise nonsynonymous and synonymous substitutions in  $TF^i$ , and  $\mathcal{N}_A(t_i)$  and  $\mathcal{N}_s(t_i)$  are the number of pairwise nonsynonymous and synonymous sites in  $TF_i$ , respectively.

#### **Supporting Information**

Figure S1 Nucleotide substitution rates at TFBSs, ChIP-chip sequences and their respective neutral control regions summed across all three species.

Found at: doi:10.1371/journal.pgen.1000157.s001 (0.43 MB EPS)

**Figure S2** Relationship between divergence (estimated summing across all three species in our study) in our neutral control regions and the expression breadth of gene in which they reside. Control divergence is not significantly correlated with expression breadth (P < 0.83).

Found at: doi:10.1371/journal.pgen.1000157.s002 (1.54 MB EPS)

Figure S3 Boxplots of TF Dn/Ds ratio by structural superclass. BSF:  $\beta$ -scaffold factors with minor groove contacts; HTH:helixturn-helix proteins; LZ:leucine zipper factors; ZC: zinc-coordinating DNA-binding domains.

Found at: doi:10.1371/journal.pgen.1000157.s003 (0.44 MB EPS)

**Figure S4** Boxplots of expression breadth of gene by structural superclass of the regulating TF. BSF:  $\beta$ -scaffold factors with minor groove contacts; HTH:helix-turn-helix proteins; LZ:leucine zipper factors;ZC: zinc-coordinating DNA-binding domains.

Found at: doi:10.1371/journal.pgen.1000157.s004 (0.48 MB EPS)

**Figure S5** TFBS divergence summed over all species plotted against year of publication of supporting literature.

Found at: doi:10.1371/journal.pgen.1000157.s005 (0.97 MB EPS)

#### Acknowledgments

We thank David Serre, Tony Kwan, Stephen Montgomery and one anonymous reviewer for useful discussions and comments on the manuscript.

#### **Author Contributions**

Conceived and designed the experiments: DJG RB. Performed the experiments: DJG RB. Analyzed the data: DJG RB. Contributed reagents/materials/analysis tools: DJG. Wrote the paper: DJG. Comments and advice on the manuscript: JM.

- Martone R, Euskirchen G, Bertone P, Hartman S, Royce TE, et al. (2003) Distribution of NF-kappa B-binding sites across human chromosome 22. Proc Natl Acad Sci USA 100: 12247–12252.
- Rinn JL, Euskirchen G, Bertone P, Martone R, Luscombe NM, et al. (2003) The transcriptional activity of human chromosome 22. Genes & Development 17: 529–540.
- Odom DT, Zizlsperger N, Gordon DB, Bell GW, Rinaldi NJ, et al. (2004) Control of pancreas and liver gene expression by HNF transcription factors. Science 303: 1378–1381.
- Cawley S, Bekiranov S, Ng HH, Kapranov P, Sekinger EA, et al. (2004) Unbiased mapping of transcription factor binding sites along human chromosomes 21 and 22 points to widespread regulation of noncoding RNAs. Cell 116: 499–509.

- Euskirchen G, Royce TE, Bertone P, Martone R, Rinn JL, et al. (2004) CREB binds to multiple loci on human chromosome 22. Molecular And Cellular Biology 24: 3804–3814.
- Kim TH, Barrera LO, Zheng M, Qu CX, Singer MA, et al. (2005) A highresolution map of active promoters in the human genome. Nature 436: 876–880.
- 12. Levy S, Hannenhalli S (2002) Identification of transcription factor binding sites in the human genome sequence. Mammalian Genome 13: 510–514.
- Liu YY, Liu XS, Wei LP, Altman RB, Batzoglou S (2004) Identification of transcription factor binding sites in the human genome sequence. Genome Research 14: 451–458.
- Dermitzakis ET, Clark AG (2002) Evolution of transcription factor binding sites in mammalian gene regulatory regions: Conservation and turnover. Molecular Biology And Evolution 19: 1114–1121.
- Sauer T, Shelest E, Wingender E (2006) Evaluating phylogenetic footprinting for human-rodent comparisons. Bioinformatics 22: 430–437.
- Mahony S, Corcoran DL, Feingold E, Benos PV (2007) Regulatory conservation of protein coding and miRNA genes in vertebrates: lessons from the opossum genome. Genome Biology 8.
- Keightley PD, Lercher MJ, Eyre-Walker A (2005) Evidence for widespread degradation of gene control regions in hominid genomes. PLoS Biol 3: e42.
- McDonald JH, Kreitman M (1991) Adaptive protein evolution at the Adh locus in Drosophila. Nature 351: 652–654.
- Smith NGC, Eyre-Walker A (2002) Adaptive protein evolution in Drosophila. Nature 415: 1022–1024.
- Clark AG, Hubisz MJ, Bustamante CD, Williamson SH, Nielsen R (2005) Ascertainment bias in studies of human genome-wide polymorphism. Genome Research 15: 1496–1502.
- Fay JC, Wyckoff GJ, Wu CI (2001) Positive and negative selection on the human genome. Genetics 158: 1227–1234.
- Khaitovich P, Hellmann I, Enard W, Nowick K, Leinweber M, et al. (2005) Parallel patterns of evolution in the genomes and transcriptomes of humans and chimpanzees. Science 309: 1850–1854.
- Duret L, Mouchiroud D (2000) Determinants of substitution rates in mammalian genes: Expression pattern affects selection intensity but not mutation rate. Molecular Biology And Evolution 17: 68–74.
- Zhang LQ, Li WH (2004) Mammalian housekeeping genes evolve more slowly than tissue-specific genes. Molecular Biology And Evolution 21: 236–239.
- Gibbs RA, Rogers J, Katze MG, Bumgarner R, Weinstock GM, et al. (2007) Evolutionary and biomedical insights from the rhesus macaque genome. Science 316: 222–234.

- 26. Saxonov S, Berg P, Brutlag DL (2006) A genome-wide analysis of CpG
- dinucleotides in the human genome distinguishes two distinct classes of promoters. Proc Natl Acad Sci USA 103: 1412–1417.27. Prabhakar S, Noonan JP, Paabo S, Rubin EM (2006) Accelerated evolution of
- conserved noncoding sequences in humans. Science 314: 786–786.
- Haygood R, Fedrigo O, Hanson B, Yokoyama KD, Wray GA (2007) Promoter regions of many neural- and nutrition-related genes have experienced positive selection during human evolution. Nature Genetics 39: 1140–1144.
- Kim SY, Pritchard JK (2007) Adaptive evolution of conserved noncoding elements in mammals. PLoS Genetics 3: 1572–1586.
- Charlesworth J, Eyre-Walker AE (2007) The other side of the nearly neutral theory, evidence of slightly advantageous back-mutations. Proc Natl Acad Sci USA 104: 16992–16997.
- Farre D, Bellora N, Mularoni L, Messeguer X, Alba MM (2007) Housekeeping genes tend to show reduced upstream sequence conservation. Genome Biology 8: R140.
- Wray GA (2007) The evolutionary significance of cis-regulatory mutations. Nature Reviews Genetics 8: 206–216.
- Gaffney DJ, Keightley PD (2006) Genomic selective constraints in murid noncoding DNA. PLoS Genetics 2: e204.
- Vinogradov AE, Anatskaya OV (2007) Organismal complexity, cell differentiation and gene expression: human over mouse. Nucleic Acids Research 35: 6350–6356.
- Keightley PD, Gaffney DJ (2003) Functional constraints and frequency of deleterious mutations in noncoding DNA of rodents. Proc Natl Acad Sci USA 100: 13402–13406.
- Webster M, Smith N, Lercher M, Ellegren H (2004) Gene expression, synteny, and local similarity in human noncoding mutation rates. Mol Biol Evol 21: 1820–1830.
- Gaffney DJ, Keightley PD (2005) The scale of mutational variation in the murid genome. Genome Res 15: 1086–1094.
- Karolchik D, Baertsch R, Diekhans M, Furey T, Hinrichs A, et al. (2003) The UCSC genome browser database. Nucleic Acids Res 31: 51–54.
- Ellegren H, Smith NGC, Webster MT (2003) Mutation rate variation in the mammalian genome. Curr Opin Genet Dev 13: 562–568.
- Yang ZH (1997) PAML: a program package for phylogenetic analysis by maximum likelihood. Computer Applications In The Biosciences 13: 555–556.